RESEARCH

Biology of Sex Differences

Open Access

Early adversity causes sex-specific deficits in perforant pathway connectivity and contextual memory in adolescent mice



Rafiad Islam¹, Jordon D. White¹, Tanzil M. Arefin^{2,3}, Sameet Mehta⁴, Xinran Liu^{5,6}, Baruh Polis¹, Lauryn Giuliano¹, Sahabuddin Ahmed¹, Christian Bowers¹, Jiangyang Zhang² and Arie Kaffman^{1*}

Abstract

Background Early life adversity impairs hippocampal development and function across diverse species. While initial evidence indicated potential variations between males and females, further research is required to validate these observations and better understand the underlying mechanisms contributing to these sex differences. Furthermore, most of the preclinical work in rodents was performed in adult males, with only few studies examining sex differences during adolescence when such differences appear more pronounced. To address these concerns, we investigated the impact of limited bedding (LB), a mouse model of early adversity, on hippocampal development in prepubescent and adolescent male and female mice.

Methods RNA sequencing, confocal microscopy, and electron microscopy were used to evaluate the impact of LB and sex on hippocampal development in prepubescent postnatal day 17 (P17) mice. Additional studies were conducted on adolescent mice aged P29-36, which included contextual fear conditioning, retrograde tracing, and ex vivo diffusion magnetic resonance imaging (dMRI).

Results More severe deficits in axonal innervation and myelination were found in the perforant pathway of prepubescent and adolescent LB males compared to LB female littermates. These sex differences were due to a failure of reelin-positive neurons located in the lateral entorhinal cortex (LEC) to innervate the dorsal hippocampus via the perforant pathway in males, but not LB females, and were strongly correlated with deficits in contextual fear conditioning.

Conclusions LB impairs the capacity of reelin-positive cells located in the LEC to project and innervate the dorsal hippocampus in LB males but not female LB littermates. Given the critical role that these projections play in supporting normal hippocampal function, a failure to establish proper connectivity between the LEC and the dorsal hippocampus provides a compelling and novel mechanism to explain the more severe deficits in myelination and contextual freezing found in adolescent LB males.

*Correspondence: Arie Kaffman arie.kaffman@yale.edu Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/.

Highlights

- The number of reelin-positive cells projecting from the lateral entorhinal cortex via the perforant pathway to the dorsal hippocampus is four times lower in adolescent male mice exposed to limited bedding (LB) compared to LB female littermates or control mice.
- Given the critical role that these projections play in hippocampal function, these findings provide a novel mechanism to explain the sex-specific deficits in contextual fear conditioning observed in adolescent LB male mice.
- Sparse axonal innervation in perforant pathway terminals also explains the reduced myelination observed in the stratum lacunosum moleculare, which is one of the most highly myelinated regions in the developing hippocampus.
- Adolescent LB male and female mice are hyperactive and exhibit significant cortical thinning, which mirrors findings in humans exposed to severe early deprivation.

Keywords Early adversity, Limited bedding and nesting, Perforant pathway, Myelination, Reelin, Cortical thinning

Plain Language Summary

Childhood adversity, such as severe deprivation and neglect, leads to structural changes in human brain development that are associated with learning deficits and behavioral difficulties. Some of the most consistent findings in individuals exposed to childhood adversity are reduced hippocampal volume and abnormal hippocampal function. This is important because the hippocampus is necessary for learning and memory, and it plays a crucial role in depression and anxiety. Although initial studies suggested more pronounced hippocampal deficits in men, additional research is needed to confirm these findings and to elucidate the mechanisms responsible for these sex differences. We found that male and female mice exposed to early impoverishment and deprivation exhibit similar structural changes to those observed in deprived children. Interestingly, adolescent male mice, but not females, display severe deficits in their ability to freeze when placed back in a box where they were previously shocked. The ability to associate "shock/danger" with a "box/place" is referred to as contextual fear conditioning and requires normal connections between the entorhinal cortex and the hippocampus. We found that these connections did not form properly in male mice exposed to impoverished conditions, but they were only minimally affected in females. These findings appear to explain why exposure to impoverished conditions impairs contextual fear conditioning in male mice but not in female mice. Additional work is needed to determine whether similar sex-specific changes in these connections are also observed in adolescents exposed to neglect and deprivation.

Background

Early-life adversity (ELA) is a heterogeneous group of childhood experiences that include abuse, neglect, severe poverty, bullying, racism, and exposure to high levels of neighborhood violence [1]. Different adversities lead to somewhat distinct developmental outcomes [2-4], which are difficult to delineate because individuals are commonly exposed to multiple adversities [4-6]. ELA is now recognized as a potent risk factor for abnormal brain development and an increased risk for multiple psychiatric and medical conditions [2, 7]. Some of the most reproducible findings in individuals exposed to ELA are reduced hippocampal volume and abnormal hippocampal function [2, 8–11]. Although some evidence indicates more pronounced volumetric changes in men exposed to ELA [2], several key questions remain unanswered. For example, it is unclear if all, or only some subtypes of ELA (e.g., abuse, neglect, etc.), lead to different structural

and functional outcomes in the hippocampus of men and women. The cellular and molecular mechanisms responsible for these sex differences and their contribution to hippocampal function and psychopathology also need to be clarified. We currently do not know whether sex differences in hippocampal function are more pronounced in childhood, adolescence, or adulthood. This is an important question, especially considering that sex differences in psychopathology and task-mediated hippocampal activation seem to be more pronounced in adolescent males [1, 9, 12].

These are difficult questions to address in the clinical setting due to the inherent heterogeneity and complexity of ELA, genetic variability, and lack of access to the developing human brain [1]. Moreover, establishing causal links between structural and functional changes in the hippocampus with alterations in cognition is difficult to achieve in humans. The conserved nature of hippocampal

development across mammalian species [13-15], coupled with the observation that ELA impairs hippocampal development and long-term function across diverse species, including rodents [16], suggests that research in rodents may elucidate important details about the underlying biology. Indeed, elegant work in rodents has identified several mechanisms by which ELA, in the form of limited bedding (LB), impairs hippocampal function in rodents [17–19]. However, most studies to date have focused on outcomes in adult males, with only few examples examining this issue in male and female rodents [16, 20, 21]. Naninck et al. [20], reported more significant hippocampus-dependent deficits in adult males than in females. They attributed these differences to reduced levels of adult neurogenesis in LB males, but others have not replicated these findings [21, 22]. The only study examining the impact of LB on hippocampal function in adolescent male and female LB mice confirmed more severe deficits in males. However, the mechanisms responsible for these sex differences have not been clarified [21].

The commonly used LB paradigm occurs from postnatal day 2 (P2) to P9 [23]. This period corresponds to developmental processes in the human hippocampus that extend from the third trimester to approximately 1-3 years of age, precluding key developmental processes seen during childhood and adolescence. These include dentate gyrus maturation, synaptogenesis, dendritic arborization, synaptic pruning, GABAergic input, and myelination [24-27]. Therefore, extending the duration of deprivation will enable a more comprehensive assessment of sexually dimorphic developmental processes occurring during childhood and particularly in adolescence [28, 29]. Finally, the work to date has focused on identifying structural and functional changes within the hippocampus without considering the impact of LB and sex on critical input from the entorhinal cortex into the hippocampus. These connections originate from reelinpositive cells in layer 2 of the lateral entorhinal cortex (LEC) and the medial entorhinal cortex (MEC) to form the perforant pathway, which is the primary input into granule cells in the dentate gyrus (DG) and, to a lesser extent, pyramidal neurons in the CA2 and CA3 subregions of the hippocampus [30]. Mature perforant pathway terminals are visible relatively early in development, around postnatal day 10 (P10) and P14 in mice and rhesus monkeys, respectively [31-33]. Functional connectivity between these terminals and granule cells in the DG undergoes significant changes during childhood and adolescence, such as myelination and expansion of synaptic input onto granule cells [27, 33]. LEC input targets newly formed neuroblasts, thereby playing an important role in guiding postnatal neurogenesis and pattern separation [34]. Disruption of perforant connectivity impairs episodic memory in rodents [30, 35–38] and children [37], underscoring an essential role for these connections in normal hippocampal development and function.

Here, we address these concerns by extending the LB procedure from P0 (birth) to P25 and testing the impact of this manipulation on hippocampal development in prepubescent (P17) and adolescent (P28-36) male and female mice. Adolescence is defined as a period of sexual maturation that starts at around P28, marked by increased expression of the neuropeptide kisspeptin, vaginal opening in females, and the onset of significant weight differences between males and females [39-43]. At around P60, mice reach sexual maturity and are considered young adults [39-43]. Hippocampal changes were investigated in prepubescent P17 mice because some aspects of hippocampal development are completed at this age (e.g., dentate gyrus formation) while other developmental processes are at their peak (e.g., synaptic pruning) or are just commencing (e.g., myelination, GABAergic innervation, and maturation of glutamatergic synapses) [24-27]. In contrast, most developmental processes in the hippocampus reach adult-like levels in P34-36 adolescent mice [44–46].

Using RNA-seq, stereology, high-resolution confocal microscopy, and electron microscopy we found that LB causes significant deficits in myelination in the hippocampus of 17-day-old prepubescent mice. Focusing on the stratum lacunosum moleculare (SLM), one of the most highly myelinated regions in the developing hippocampus, we demonstrate that decreased myelination results from reduced axonal innervation. These deficiencies were more prominent in prepubescent males and persisted into adolescence. High-resolution ex vivo diffusion magnetic resonance imaging (dMRI) in P29 adolescent mice revealed significant cortical atrophy and reduced hippocampal volume in both male and female mice. The volume of the entorhinal cortex and structural connectivity with the dorsal hippocampus were more prominently reduced in 29-day-old adolescent LB males compared to LB females. Using retrograde tracing, we found that LB impairs the ability of reelin-positive cells in the lateral entorhinal cortex (LEC) to project and connect with the dorsal hippocampus in 36-day-old adolescent LB male mice, a reduction that was not observed in female LB littermates. Finally, deficits in connectivity between the LEC and the dorsal hippocampus were strongly correlated with abnormal contextual fear conditioning. These results suggest that reduced connectivity between the LEC and the dorsal hippocampus contributes to abnormal contextual fear conditioning in adolescent LB male mice. Female LB littermates, on the other hand, are able to maintain normal connectivity and intact contextual fear conditioning.

Methods

Animals

BALB/cByj mice (Jackson laboratories, stock # 001026) were bred in-house and kept on a standard 12:12 h light– dark cycle (lights on at 7:00 AM), with food provided ad libitum. Temperature and humidity were held constant ($23 \pm 1 \degree$ C and $43\% \pm 2$), and background noise in the room was kept at dB: 56.5. All studies were approved by the Institutional Animal Care and Use Committee (IACUC) at Yale University and were conducted in accordance with the recommendations of the NIH Guide for the Care and Use of Laboratory Animals.

Limited bedding (LB)

The limited bedding (LB) procedure was performed as described previously [47, 48]. Briefly, breeding cages were set up using a 3:1 female-to-male harem in standard mouse Plexiglas cages with 2 cups of corncob bedding and no nesting material. Visibly pregnant dams were transferred to maternity cages containing 2 cups of corncob bedding with no nesting material and 3 chow pellets on the floor. At birth (P0), litters were culled to 5-8 pups and randomized to either control (CTL) or limited bedding (LB). CTL litters were provided with 500 cc of corncob bedding, 15 cc of soiled bedding from the birth cage, and one 5×5 cm nestlet from P0-25. LB litters were provided with 125 cc corncob, 15 cc of soiled bedding from the birth cage, and no nestlet from P0-25. Bedding was changed on P7, P14, and P21. All mice were weaned on P26 and housed with 2-3 same-sex and condition littermates per cage with 500 cc of corncob bedding, no nesting material, and 2-3 chow pellets on the floor.

Tissue collection and experimental design

Behavioral studies were conducted between (9:30–12:30) and tissue was collected between (13:00-15:00) to minimize changes associated with circadian rhythm [47, 48]. Prepubescent mice (n=117) were processed at P17 for RNA-seq, stereology, biochemical assessment of DNA, RNA and protein contents in the hippocampus, immunohistochemistry, and electron microscopy (Additional file 1: Fig S1A). Exploratory behavior in a group of P29 adolescent mice (n=69) was examined using the open field test with a subset of the mice immediately perfused after the behavioral testing for dMRI studies (n=24, Additional file 1: Fig S1B). A second group of adolescent mice was tested in the contextual fear conditioning at P31-33 (n=66) and then perfused for immunohistochemistry at P34 (n=24, Additional file 1: Fig S1C). Finally, a third group of adolescent mice was injected with the retrograde tracer CTB at P28, tested in the contextual fear conditioning at P33-34 (n=46) and perfused to assess the number of reelin-positive and CTB-positive cells in the lateral entorhinal cortex at P35 (n=38, Additional file 1: Fig S1D).

Assessing RNA, DNA and protein contents in the P17 hippocampus

To characterize RNA, DNA, and protein content in the hippocampus, P17 pups were anesthetized, and the right and left hippocampi were dissected, snap-frozen in liquid nitrogen, and stored at - 80 °C until further processing (n=8-9 mice per rearing and sex). The left hippocampus was homogenized on ice in lysis buffer from the AllPrep® DNA/RNA/Protein Mini Kit (QIAGEN, Hilden, Germany). A sample from the lysate was used to quantify total DNA and RNA content using the appropriate Qubit® Fluorometer kit (Cat. # Q32850 for DNA and Q32852 for RNA). The remaining homogenate was then processed according to the manufacturer's instructions to purify RNA, DNA, and proteins. Samples were stored at -80 °C until further processing. Purified total protein was quantified using a PierceTM BCA Protein Assay Kit (Cat. # 23,227).

RNA-seq

RNA samples from the left hippocampus of P17 pups (CTL males=7, CTL females=8, LB males=8, LB females = 8 from 4-6 independent litters) were sequenced at the Yale Center for Genomic Analysis. RNA quality and concentration were assessed using the 2100 Bioanalyzer on an RNA 6000 Nano Assay (Agilent Technologies, Inc., all RINs > 9). Libraries were constructed by amplifying 500 ng of total RNA using 9 PCR cycles with the KAPA mRNA HyperPrep Kit (KAPA Biosystems). cDNA libraries were validated using the Bioanalyzer 2100 on a High Sensitivity DNA assay and quantified using the KAPA Library Quantification Kit for Illumina Platforms on a Roche Lightcycler 480. Sequencing was performed on an Illumina NovaSeq 6000 using the S4 XP workflow 2×100 with each library using 1.25% of the S4 lane. Reads were trimmed for quality using custom scripts with a minimum accepted length of 45 bases. The trimmed reads were then aligned to the mm10 reference mouse genome using gencode annotation [49] and HISAT2 software [50]. Transcripts per million (TPM) estimation was calculated using StringTie [50].

Immunohistochemistry

Perfused brains were sectioned coronally using a VT1000S vibratome (Leica) to obtain 50-micron sections and sorted into 6 pools, each containing 16–18 slices spaced at 300-micron intervals that systematically span the entire rostral-caudal axis of the brain. No staining was seen when primary antibodies were

omitted, and the specificity of all antibodies was previously established [51-54] and further confirmed using RNA-seq and electron microscopy. Tissue was blocked with TBST buffer for one hour at room temperature [1X Tris-buffer saline (Bio-Rad, cat#1,706,435), 0.3% Triton X100 (Sigma-Aldrich, Cas# 9036-19-5), and 10% Normal goat serum (Jackson ImmunoResearch Laboratories Inc. cat# 005–000-121, RRID: AB 23369900]. After blocking, free-floating sections were incubated overnight at room temperature with rabbit PDGFRα (D1E1E) XP[®] antibodies (Cell Signaling, cat #3174, RRID:AB 2162345, 1:200), mouse anti-APC (Ab-7) (also known as anti CC-1, Millipore, cat #OP80-100UG, RRID:AB 2057371,1:100), and rat anti-MBP (Bio-Rad, cat # MCA409S, RRID:AB 325004, 1:500). Axonal staining and myelination in the stratum lacunosom moleculare (SLM) were assessed with rat anti-MBP (Bio-Rad, cat # MCA409S, RRID:AB 325004, 1:500), mouse anti-neurofilament H (NF-H) phosphorylated antibodies (BioLegend, Cat# 801,602, RRID:AB_2715851, 1:1000). The number of reelin-positive cells was assessed using a mouse anti-reelin antibody (Abcam, Cat # ab78540, RRID:AB 1603148, 1:400). The slices were then incubated with the appropriate fluorescently labeled goat secondary antibodies (Thermo Fisher, Cat #A-21422/ RRID:AB 141822, # A-11008/RRID:AB 143165, # A-21094/ RRID:AB 141553, all at 1:200) and mounted on glass slides with VECTASHIELD Hard-Set antifade mounting medium with DAPI (Vector laboratories Cat# 10,955).

Stereology, cell counting, and image analysis

Stereological analysis was performed by tracing the borders of the granule cell layer of the dentate gyrus (DG) and the pyramidal cell layer of the Cornu Ammonis (CA) regions of the hippocampus under low magnification (-2.5x) using Stereo Investigator 10 software (MBF Bioscience), and volume was calculated using Cavalieri's principle [55, 56] and normalized for body weight. To quantify PDGFRα and CC1 cell numbers and MBP intensity, a $450 \times 250 \,\mu\text{m}$ square area containing the SLM was selected and analyzed using Imaris 10.0 (Bitplane, Oxford Instruments). The object selection spots model was used to analyze PDGFRa and CC1 with the cell diameter set to 6.5 μ m for PDGFR α and 13.5 μ m for CC1, and the Z diameter was kept as default. MBP and NF-H intensities were determined using the object selection surface model using a 0.720 μ m threshold with the intensity max filter applied. Colocalization between MBP and NF-H was performed with the Object-Based-Colocalization model using the shortest distance to surfaces of 0.1 μ m. Three slices were counted per animal and averaged to determine the number of PDGFRa- and CC1-positive cells, MBP intensity, and MBP/NF-H colocalization for

each mouse. For counting CTB-positive cells, a 200 X 300 μm square area containing layers 2–3 of the LEC was cropped, and the number of cells was counted in Imaris using the Spots Model with XY diameter of 20 μm , default Z-diameter, and max intensity filter settings. Five slices were counted per animal and averaged to obtain the number of reelin-positive cells in the LEC for each animal.

Electron microscopy

P17 male mice were perfused with cold PBS followed by 4% paraformaldehyde in PBS. Brains were then dissected out and immersed in a solution containing 2.5% glutaraldehyde and 2% paraformaldehyde in 0.1 M sodium cacodylate buffer (pH 7.4) and kept overnight at 4 °C. Two hundred micron-thick sections containing the dorsal hippocampus (Paxinos coordinates, bregma = -1.34 mm to - 2.80 mm) were cut using a VT1000S vibratome (Leica) and postfixed with 1% OsO4 and 0.8% potassium ferricyanide at room temperature for one hr. Specimens were then en bloc stained with 2% aqueous uranyl acetate for 30 min, dehydrated in a graded series of ethanol to 100%, substituted with propylene oxide, and embedded in EMbed 812 resin. Sample blocks were polymerized in an oven at 60 °C overnight. To locate SLM, semithin sections (250 nm) were prepared and stained with a solution of toluidine blue and 1% sodium borate. TEM thin sections (60 nm) were obtained using a Leica ultramicrotome (UC7) and poststained with 2% uranyl acetate and lead citrate. Sections were examined with an FEI Tecnai transmission electron microscope at an accelerating voltage of 80 kV, and digital images were recorded with an Olympus Morada CCD camera and iTEM imaging software. The MyelTracer software [57] was used to determine axonal diameter and G-ratio. This software automatically converts various axonal shapes into perfect circles and calculates the G-ratio as the ratio between the inner axonal diameter and the diameter of the axon plus the myelin sheath that encases it. N = 20 axons per mouse and n = 3 male mice per group.

Behavior

Open field test was done as previously described [47, 58]. Briefly, mice were allowed to explore a 40×40 cm arena (Maze Engineers, # 3201) illuminated at $60 \, l \times$ for 5 min during which the distance traveled and the time spent in the inner 15 cm-area were measured using the EthoVision tracking system (XT 17, Noldus Information Technology). Contextual fear conditioning was performed as previously described [58], except that acoustic cue conditioning was not tested on the third day of contextual fear conditioning. Briefly, during the first day of training, mice were allowed to explore a Med Associates' fear

conditioning chamber (Cat # VFC-008) equipped with a grid floor in the presence of a 2% lemon scent for 300 s. After 300 s of free exploration, mice were exposed to 5 shock-tone pairing at variable inter-trial interval (30–180 s). These were presented as 30 s discontinued tones (7500 Hz, 80 dB) that co-terminated with a 1 s 0.65 mA foot shock. There was a 30 s period at the end of the session following the last pairing and freezing behavior recorded using Med Associates' Video Freeze software. On the second day, mice were returned to the same box and contextual freezing was determined for 300 s in the absence of a tone or a shock.

Ex vivo dMRI

P29 adolescent CTL and LB mice (N=6 mice per rearing and sex, a total of 24 mice) were perfused and processed for high-resolution ex vivo dMRI using a 7-Tesla MR system equipped with a 4-channel cryogenic probe as described previously [58, 59]. Whole-brain voxel-based morphometric analysis was used to identify local volumetric changes affected by rearing, sex, and their interaction (2×2 ANOVA, FDR corrected, α =0.1, p<0.0105, cluster size>25 voxels). Fiber tractography was used to assess structural connectivity between the entorhinal cortex and the dorsal hippocampus following a previously described methodology [58–60].

Retrograde labeling

P29 mice were anesthetized with isoflurane (1–3%), and 50 nl of the retrograde tracer Alexa Fluor[™] 555 cholera toxin subunit B (Thermo Fisher Scientific, Cat # C34776, 0.5% wt./vol in PBS) was injected at 10 nl/sec into the left SLM (ML: – 1.2, AP: – 2, DV: – 1.9) using a 5 µL Hamilton syringe equipped with a 32 blunt needle (Hamilton, Cat# 7803–04, 0.5 inch, PST 3). After injection, the needle was left in place for 5 min and then slowly removed. The incision was closed with Vet Bond, and the mice were allowed to fully recover prior to being returned to their home cage.

Statistical analysis

Statistical analyses were performed using MATLAB R2022b (www.mathworks.com), SPSS (Version 29.0. Armonk, NY: IBM Corp.) and visualized with GraphPad Prism 9.0 (GraphPad Software, La Jolla California USA). The data were carefully screened for inaccuracies, with normality and homogeneity of variance confirmed using Kolomogrov-Smirnov and Mauchly's tests, respectively. Outliers were removed if they were more than 2 standard deviations above or below the mean. Sample sizes were determined based on effect sizes obtained from preliminary studies, with alpha=0.05, and a power >0.8. Data were examined using a 2×2 ANOVA with rearing

condition (CTL, LB) and sex as fixed factors. Significant rearing by sex interaction (e.g., p < 0.05) was followed by Sidak's post hoc analysis for each sex. Significant main effect of rearing or interaction were followed by a preplanned quantification of Cohen's simple effect sizes for each sex. RNA-seq analysis was conducted by first removing genes with low expression (average TPM < 3 in all three groups). This was done to increase statistical power and focus on genes that are more likely to drive developmental changes. Counts of the 13,245 genes that passed the initial screen were analyzed using DESeq2 script in R [61] to identify genes that were differentially expressed between CTL and LB using a Benjamini-Hochberg false discovery rate (FDR) < 0.05. All raw data and analyses are available at GEO archive accession # GSE142305. Enrichment and transcription factor analyses were conducted using MetaCore[™] (Clarivate Analytics) workflows on gene lists that passed a corrected p value of < 0.05.

Results

LB Inhibits oligodendrocyte progenitor-cell differentiation in the stratum lacunosom moleculare (SLM)

Consistent with previous studies [47, 48, 58], LB decreased body weight in P17 male and female pups (Additional file 1: Fig S2 A, B). Stereological analysis indicated that LB decreased the size of the granule cell layer (GCL) and the Cornu Ammonis (CA) pyramidal layer even after correction for body weight, with similar outcomes observed in males and females (Additional file 1: Fig S2C). The DNA, RNA, and protein contents extracted from the hippocampus were also reduced in male and female P17 LB pups (Additional file 1: Fig S2D). These findings replicate previous research demonstrating that LB reduces hippocampal volume in prepubescent pups [22, 47].

Bulk RNA-seq was used to further elucidate the mechanisms by which LB alters hippocampal development (Fig. 1A). Similar changes in gene expression were observed in male and female LB mice prompting us to focus on rearing-induced changes in gene expression. Using this approach, we identified 797 differentially regulated genes (DRGs) between CTL and LB (FDR < 0.05), most of which had a small effect size (i.e., $\log 2 < 0.4$, green/turquoise heatmap, Fig. 1B). Pathway analysis identified oligodendrocyte differentiation and myelination as the most significantly downregulated pathways in LB, whereas cytoskeleton remodeling was the most upregulated pathway in LB (Fig. 1C). The 20 most upregulated and downregulated genes are shown in Additional file 1: Fig S2. Transcription factor analysis identified reduced activity of Myrf, Olig2, and Sox10 as the key regulators of abnormal myelination in LB mice (Fig. 1D). Since



Fig. 1 LB inhibits oligodendrocyte differentiation in the hippocampus of P17 prepubescent pups. **A** Experimental timeline. **B** Heatmap showing the effect size of differentially regulated genes (DRGs) at FDR < 0.05 between CTL and LB. **C** Pathway analysis of the top downregulated and upregulated pathways affected by LB. **D** Transcription factor analysis identified changes in Myrf, Olig2 and Sox10 activity as the most dysregulated factors affected by LB. LB does not affect the expression of OPC-specific genes **E** but significantly decreases the expression of genes expressed in mature oligodendrocytes (**F**–**G**). N=7–8 mice per rearing and sex group. Error bars represent mean \pm SEM. *p < 0.05, **p < 0.01, ****p < 0.001

Myrf is necessary for oligodendrocyte differentiation [53, 62], we assessed the effects of LB on genes that are specifically expressed in oligodendrocyte progenitor cells (OPCs) and mature oligodendrocytes (OLs). LB had no impact on OPC-specific genes (Fig. 1E) but significantly reduced the expression of OL-specific genes (Fig. 1F), indicating that LB impairs the differentiation of OPCs in the developing hippocampus. A 2×2 ANOVA examining the impact of rearing and sex on some of the most highly expressed OL-specific genes confirmed a highly significant effect of rearing (Fig. 1G). No significant effect of sex or interaction were found, except for Plp1 mRNA, where a significant effect of sex was observed. Nevertheless, a preplanned analyses of effect sizes found greater deficits in LB males (Fig. 1G. Cohen's d effect sizes MBP: males = 2.45 vs females = 1.3, CNP: males = 2.71 vs females = 1.22, MOBP: males = 2.61 vs females = 1.21, Plp1: males = 2.66 vs females = 1.28).

To further validate the RNA-seq data, we assessed the effects of LB on oligodendrocyte maturation in the stratum lacunosum moleculare (SLM), one of the

most myelinated regions in the developing hippocampus (Fig. 2A) [27, 63, 64]. This was done by quantifying levels of MBP as a marker of myelination, number of PDGFR α -positive cells (OPCs), and number of CC1 positive cells (OL) in the SLM of P17 pups. Consistent with the RNA-seq data, we found a significant reduction in MBP staining in LB compared to control (F (1, 16)=15.42, P=0.0012, η_p^2 =0.49) with no significant effects of sex or interaction (Fig. 2B, C). There was also a significant reduction in the number of PDGFRa-positive cells in LB pups (Fig. 2B, D, rearing: F (1, 16)=5.88, P = 0.0275, $\eta_p^2 = 0.27$, sex: F (1, 16) = 0.05, P = 0.8, interaction: F (1, 16) = 0.01, P=0.91). For CC1-positive cells (i.e., mature-OL), there was a significant effect of rearing (F (1, 16)=35.56, P<0.0001, η_p^2 =0.69) and sex (F (1, 16) = 4.664 P = 0.046). No significant interaction between rearing and sex was found but LB males showed a significantly greater reduction compared to LB female littermates (Cohen's d effect size males = 4.58, females = 1.67, Fig. 2E). Since LB reduced both OPC and OL in the SLM, an OPC differentiation index was generated by



Fig. 2 OPC differentiation in the SLM is impaired in P17 Prepubescent LB mice. **A** Low (4X) and higher (20X) magnification of myelin basic protein (MBP) staining in the SLM of P17 pups. **B** Representative images of PDGFRa-positive OPCs, CC1-positive mature oligodendrocytes, and MBP in the SLM. Quantification of MBP (**C**), PDGFRa-positive OPCs (**D**), CC1 mature oligodendrocytes (**E**) and OPC differentiation index (**F**) in the SLM group. N=5-6 mice per rearing and sex group. Error bars represent mean \pm SEM. *p<0.05, **p<0.01, ****p<0.001

dividing the number of CC1-positive cells by the number of PDGFR α -positive cells. There was a significant reduction in the OPC differentiation index in LB mice (F (1, 16)=32.4, P<0.000, η_p^2 =0.67, Fig. 2F) indicating that impaired OPC differentiation is primarily responsible for the reduced myelination in the SLM. Again, no significant effect of sex or interaction were found, but males were more severely impacted compared to females (Cohen's d effect size males=3.99, females=1.50, Fig. 2F).

Axonal staining is reduced in the SLM of P17 LB mice

To test whether the reduced myelination is due to abnormal axonal innervation, we probed the hippocampus with anti-phosphorylated neurofilament H antibodies (NF-H) to quantify axonal fibers in the SLM of P17 control and LB mice (Fig. 3A). There was a significant reduction in NF-H staining in LB mice (F (1, 16) = 11.38, P = 0.0039), with no significant effect of sex and a trend for rearing by sex interaction (F (1, 16) = 3.30, P = 0.088). The trend in interaction was due to a greater reduction in NF-H staining in LB males compared to LB females (Fig. 3B, Cohen's d effect size males = 2.05, females = 0.82). A similar pattern was observed for MBP staining, with reduced myelination in LB males (P < 0.0001, Cohen's d = 3.23) but not in females (P=0.48, Cohen's d=1.11, interaction: F (1, 16) = 14.20, P = 0.0017, Fig. 3C). The percentage of neurofilaments co-stained with MBP was reduced in LB compared to CTL (F (1, 16) = 5.78, P = 0.029), with no significant effects of sex or interaction, but a larger impact in males compared to females (Cohen's d effect size males = 2.28, females = 0.45, Fig. 3D). These results suggest that reduced axonal availability contributes to the abnormal myelination observed in the SLM of 17-day-old prepubescent LB mice, with males being more impacted than females.

Since LB prepubescent males show more pronounced deficits in myelination, we used electron microscopy to further characterize myelination in the SLM of P17 LB males. This investigation confirmed reduced axonal myelination in the SLM of 17-day-old LB males (Fig. 4A). LB males exhibited a significant increase in the G-ratio (t (118)=15.57, P<0.0001, Cohen's d = 2.84, Fig. 4B). Axonal diameter was also increased in LB compared to CTL males (CTL = $1.05 \pm 0.04 \mu m$, $LB = 1.47 \pm 0.074 \ \mu m$, t (95) = 4.87, P < 0.001, Cohen's d = 0.89), indicating that the reduced myelination was not due to reduced axonal diameter. Furthermore, the increase in the G-ratio observed in LB remained significant even after adjusting for the increase in axonal diameter (ANCOVA, F=130.759, P<0.001, Fig. 4C). This indicates that the elevated G-ratio seen in LB males is mainly influenced by decreased myelination.



Fig. 3 LB causes a more pronounced reduction in axonal staining in the SLM of Prepubescent 17-day-old LB male mice. A Representative images of axonal staining using anti-NF-H and anti-MBP antibodies in the SLM. Quantification of NF-H staining (**B**), MBP staining (**C**) and NF-H-MBP colocalization (**D**). Scale bars in Fig. 4A are 50 microns. N=5 mice per rearing and sex. Error bars represent the mean \pm SEM. *p < 0.05, **p < 0.01, ****p < 0.0001

Adolescent P29 LB mice are hyperactive and show extensive cortical atrophy

We have recently proposed that LB is a mouse model of childhood deprivation and neglect [1, 48]. Given that hyperactivity and extensive cortical thinning are consistent findings in children and adolescents exposed to severe social and cognitive deprivation [65, 66], we used the open field test and high-resolution dMRI to evaluate these two outcomes in P29 adolescent male and female mice (Fig. 5A). A 2×2 ANOVA for body weight revealed a significant effect of sex (F (1, 67) = 14.46, P = 0.0003), with males being larger than females (Fig. 5B), consistent with previous research indicating that puberty initiated at around this age in mice [39–43]. No significant effects of rearing or interaction were observed for body weight at this age (Fig. 5B). LB male and female P29 adolescent mice showed a significant increase in total distance traveled in the open field test (F (1, 65) = 3.95, P = 0.05), with no significant effect of sex or interaction (Fig. 5C). There was a significant interaction between rearing and sex for the time spent in the center (interaction: F (1, 65) = 7.71, P = 0.0072, Fig. 5D) and frequency of entering the center (interaction: F (1, 71)=3.92, P=0.05, Fig. 5E) due to an increased entry into the center in LB females but not LB males. These findings replicate our previous research in adolescent LB mice [47] and confirm increased hyperactivity in LB adolescent male and female mice.

Next, we conducted a 2×2 whole-brain voxel-based analysis to identify volumetric changes induced by rearing, sex, and their interaction [58, 59]. This unbiased approach revealed extensive volumetric reduction in LB adolescent male and female mice that included the sensory and motor cortex, prefrontal cortex, hippocampus, amygdala, and thalamus (shown in red in Fig. 5F). Followup region of interest analysis confirmed reduced CA1 (F (1, 20) = 15.25, P = 0.0009) and dentate gyrus volumes (F (1, 20) = 13.37, P=0.0016) in both LB male and female P29 mice (Fig. 5G). These findings indicate that the lower hippocampal volume, seen in P17 LB pups (Additional file 1: Fig S2), persists in adolescent mice and is likely to represent global atrophy across multiple cortical and subcortical gray matter regions. Few brain regions showed a significant effect of sex (Additional file 1: Fig S4), and the only brain region that showed significant rearing by



Fig. 4 Transmission electron microscopy of myelinated axons in the SLM of P17 Prepubescent CTL and LB male mice. **A** Left, a light microscopy image of the same toluidine-stained sample to identify the SLM; Right, TEM images of the SLM in CTL and LB mice (visible axons indicated with red dots). **B** Image analysis of covariance between the G-ratio and axonal diameter (**C**). N = 20 axons per mouse with 3 male mice per condition. Error bars represent the mean \pm SEM. ****p < 0.0001

sex interaction was the left entorhinal cortex (Fig. 5H). A region of interest analysis confirmed a significant sexby-rearing interaction in the left entorhinal cortex (F (1, 20)=4.397, P=0.049) driven by reduced volume in LB males (P=0.0015, Cohen's d=2.18) but not LB females (P=0.54, Cohen's d=0.61, Fig. 5I). A similar analysis in the right entorhinal cortex found a significant rearing effect (F (1, 20)=11.85, P=0.0026). No significant impact of sex (F (1, 20)=0.61, P=0.44) or interaction (F (1, 20)=1.39, P=0.25) were found, but the effect size in males was again larger (Cohen's d effect size males=1.72, females=1.03, Fig. 5J).

Reduced axonal innervation persists in P33 adolescent mice and correlates with deficits in contextual fear conditioning

Next, we tested whether the reduction in myelination and axonal staining seen in the SLM of P17 pups persisted and contributed to hippocampus-dependent deficits in adolescent mice. This was done by first testing contextual freezing in P31-33 adolescent mice and then assessing axonal staining and myelination in the SLM (Fig. 6A). Mice showed the expected increase in freezing in response to shock on the first day of training (F (2.958, 136.1) = 208, P < 0.0001). However, there was a significant interaction between rearing and sex for contextual freezing during the second day (F (1, (62) = 3.994, P = 0.05), driven by a significant reduction in freezing behavior in LB males (P = 0.0008, Cohen's d = 1.55) but not LB females (P = 0.43, Cohen's d = 0.36, Fig. 6B). Examination of NF-H staining in the SLM revealed significant effects of rearing (F (1, 20) = 9.55, P = 0.0058) and sex (F (1, 20) = 4.46, P = 0.047) but no significant interaction (F (1, 20)=0.426, P=0.52) (Fig. 6C). Nevertheless, as with the axonal staining at P17 (Fig. 4), the reduction in axonal staining was more pronounced in LB males (Cohen's d=2.01) than in LB females (Cohen's d = 1.27), and the intensity of staining was significantly correlated with contextual freezing (Fig. 6E). No significant effects of rearing, sex, or interaction were observed for the number of PDGFR1a progenitor cells or CC1 mature oligodendrocytes in the SLM (Additional file 1: Fig S5A-C). However, there was a significant reduction in MBP staining in the SLM of LB adolescent mice (F (1, 19) = 4.40, P = 0.049) that was again more pronounced in males compared to females (Additional file 1: Fig S5D). Together, these results indicate that reduced axonal staining and myelination persist in the SLM of adolescent LB mice and may contribute to the sex-specific deficits seen in contextual fear conditioning.



Fig. 5 Adolescent P29 LB mice are hyperactive and show extensive reduction in cortical and subcortical gray matter. **A** Experimental timeline. Effects of rearing and sex on body weight (**B**), and exploratory behavior in the open filed test in P29 adolescent mice (**C**–**E**). CTL males N=15, CTL females N=14, LB males N=16, LB females N=24. **F** High resolution ex vivo dMRI found significant effects of rearing on volumetric changes using (minimal cluster size > 25 voxels, FDR < 0.1, p < 0.005). Areas with significant volume reduction are shown in red. N=6 per rearing and sex. **G** Effects of rearing and sex on CA1 and dentate gyrus volumes. **H–J** The left entorhinal cortex was the only region showing a significant rearing by sex interaction (minimal cluster size > 5 voxels, FDR < 0.1, p < 0.0023). Al: Insular cortex, CB: Cerebellum, cpd: Cerebral peduncle, ENT: Entorhinal cortex, GP: Globus pallidus, HB: Hindbrain, HPF: Hippocampus, HY: HY: Hypothalamus, Int: Internal capsule, LSX: Lateral septal complex, MO: Motor cortex, OLF: Olfactory area, opt: Optic tract, OT: Olfactory tubercle, PFC: Prefrontal cortex, PTLp: Posterior parietal association cortex, RSP: Retrosplenial cortex, SC: Superior colliculus, SS: Somatosensory cortex, STR: Striatum, TH Thalamus, VIS: Visual cortex

LB reduces reelin-positive projections in male mice

Most axonal terminals in the SLM originate from the entorhinal cortex [27], suggesting that reduced axonal staining in the SLM (Fig. 6D) reflects abnormal connectivity between these two brain regions in LB adolescent mice. Indeed, dMRI tractography revealed a significant reduction in structural connectivity between the entorhinal cortex and the dorsal hippocampus on the left (F (1, 20)=8.241, P=0.0095, Fig. 7A, B) and the right hemispheres (F (1, 20)=5.659, P=0.0274, Fig. 7C). Although there were no significant effects of sex or interaction between rearing and sex, the effect sizes in males were larger than those in females (Cohen's d left: M=1.39, F=0.92; Cohen's d right: M=1.65, F=0.39).

To further validate these findings, we administered a fluorescently labeled retrograde tracer, Alexa-555-CTB,

into the left SLM of P28 CTL and LB mice, tested contextual fear conditioning at P34, and processed the mice a day later to assess the number of CTB-positive cells in the left entorhinal cortex (Fig. 8A). LB mice showed a significant reduction in contextual freezing (F (1, 23)=6.80, P=0.016) that was more pronounced in males than in females (Fig. 8B). No significant effects of rearing, sex or interaction were observed for CTB labeling at the injection site (Additional file 1: Fig S6). However, there was a significant interaction between rearing and sex for the number of CTB-positive cells in the LEC (F (1, 12)=11.08, P=0.0060). This interaction was due to a fourfold reduction in the number of CTB-positive cells in LB males (P<0.0001, Cohen's d=3.69) that was not observed in LB females (P=0.22,



Fig. 6 LB causes more severe deficits in contextual fear conditioning in adolescent P34 male mice. A Experimental timeline. B Contextual fear conditioning. CTL male = 17, CTL females = 18, LB males = 13, LB females = 18. C, D NF-H axonal staining in the SLM of adolescent mice, N = 6 mice per rearing and sex group. E The intensity of axonal staining correlates with contextual freezing behavior. Error bars represent the mean \pm SEM. *p < 0.05, **p < 0.01, ***p < 0.001

Cohen's d = 1.69, Fig. 8C, D). Moreover, the number of CTB-positive cells in the LEC was highly correlated with freezing behavior (Fig. 8E). Together, these findings indicate that LB reduces connectivity between the entorhinal cortex and the dorsal hippocampus in a manner that correlates with deficits in contextual memory and is more pronounced in adolescent males.

Previous work has shown that reelin-positive cells project from the LEC to the dorsal hippocampus and that these projections are essential for contextual memory [30, 35, 36]. To determine whether the CTB-positive cells in the LEC were reelin-positive, we repeated the CTB retrograde labeling described in Fig. 8A using a second cohort of adolescent mice and replicated the sex-specific reduction in the number of CTB-positive cells in the LEC (Interaction: F (1, 13) = 10.76, P = 0.0060, Fig. 9A, B). We confirmed that all CTB-positive cells were also reelin-positive, regardless of rearing and sex (Fig. 9A, C), and found that the total number of reelin-positive cells in the LEC was reduced in LB mice, an effect that was more pronounced in males (rearing: F(1, 13) = 31.39, P<0.0001, sex: (F (1, 13)=25.50, P=0.0002, interaction: (F (1, 13)=3.399, P=0.088, Fig. 9A, D). The percentage of reelin-positive and CTB-negative cells was significantly higher in LB males than in all other groups (interaction: F (1, 13) = 6.375, P=0.025, Cohen's d effect size males = 2.63, females = 1.19, Fig. 9A, E), indicating that most reelin-positive cells in the LEC fail to project to the dorsal hippocampus in LB males.

Discussion

Clinical and preclinical studies have vielded evidence suggesting that exposure to ELA affects hippocampal development and function differently in males and females [1, 2, 9, 20, 21]. However, these findings necessitate additional replication and a more comprehensive exploration of potential underlying mechanisms. Consistent with previous work, we found robust deficits in contextual fear conditioning in adolescent males but not female mice. Exposure to LB reduced hippocampal volume in both males and females (Fig. 5F-G), suggesting that volumetric changes are unlikely to explain the sex differences in contextual freezing. In contrast, LB caused sex-specific volumetric changes in the entorhinal cortex and its connectivity with the dorsal hippocampus via the perforant pathway. More specifically, LB adolescent males showed a fourfold reduction in the number of reelin-positive projections to the dorsal hippocampus compared to CTL mice and LB females (Figs. 8D, 9B). Approximately 60% of the reelin-positive cells in the LEC of LB males fail to project to the dorsal hippocampus (Fig. 9E), suggesting that abnormal axonal pathfinding



Fig. 7 LB reduces structural connectivity between the entorhinal cortex (ENT) and the dorsal hippocampus (dHPC) in P29 mice. Representative images (**A**) and quantification of ENT-dHPC tractography in the left (**B**) and right (**C**) hemispheres. N = 6 mice per rearing and sex condition. Error bars represent the mean \pm SEM. *p < 0.05, **p < 0.01

and/or axonal degeneration/retraction [67-69] are responsible for the reduced perforant pathway connectivity in LB male mice. The critical role that these projections play in contextual memory [30, 35-38] and the significant positive correlation between the number of these projections and contextual freezing suggest a compelling and novel mechanism to explain the more severe hippocampus-dependent cognitive deficits in LB adolescent males. More severe deficits in axonal staining and myelination were also observed in P17 prepubescent LB male pups (Figs. 1-3) suggesting that these sex differences emerge during the 2nd week of life when connectivity between the LEC and the dorsal hippocampus matures [27, 33]. Further studies are needed to clarify the exact age and the mechanisms by which LB impairs the formation of these connections. Additional studies are also needed to clarify the developmental impact of changes in connectivity on neurogenesis, synaptogenesis, and myelination and whether strategies that augment these connections can normalize contextual memories in LB male mice. Whether reduced perforant pathway connectivity alters synaptic maturation is of particular interest given recent work showing that LB reduced synaptic density and local functional connectivity in the dorsal hippocampus of adolescent males, but not LB female littermates [70].

LB replicates key features of childhood deprivation in humans

Childhood neglect/deprivation is the most common form of early adversity [71, 72] leading to structural, behavioral, and cognitive deficits that are not typically seen in other forms of ELA, such as physical and sexual abuse [73–75]. Some key features of childhood neglect



Fig. 8 LB causes sex-specific deficits in lateral perforant pathway connectivity in P35 adolescent mice. A CTL and LB mice were injected with the retrograde tracer Alexa 555-CTB into the left SLM at P28, tested in contextual fear conditioning (P33-34), and perfused to assess the number of CTB-positive cells in the LEC at P35. B Contextual fear conditioning. C Schematic depiction of CTB injection at P28 and CTB labeling in the LEC at P35. D Number of CTB-positive cells in the LEC at P35. E The number of CTB-positive cells in the



Fig. 9 LB reduces the number of reelin-positive projections in adolescent males. **A** Representative images of Reelin- and CTB-positive cells in the LEC (scale bars in **A** are 50 μm). Higher magnification of rectangular areas of merged images is shown on the right. **B** The number of CTB-positive cells is reduced in the LEC of male LB mice. **C** All CTB-positive cells are reelin-positive regardless of rearing and sex conditions. **D** LB reduces the number of reelin-positive cells in the LEC. **E** Effects of rearing and sex on the percentage of reelin-positive CTB-negative cells in the LEC.

and deprivation include hyperactivity and cortical thinning [65, 66, 73–77], outcomes that have not yet been described in rodent models of ELA. The hyperactivity and extensive cortical atrophy observed in adolescent LB mice suggest that LB is a mouse model of early deprivation. These findings were seen in both males and females, indicating that sex-specific sequelae are not a global phenomenon but rather circuit specific.

Childhood neglect causes significant deficits in declarative memory [76, 77]. However, it is currently unclear whether these hippocampal deficits are more pronounced in adolescent males and whether they are associated with abnormal perforant pathway connectivity. Our ability to use imaging to quantify these connections and to assess their contribution to episodic memory in humans and mice provides an important strategy to validate findings across species. For example, exposure to traumatic brain injury in childhood disrupts entorhinal cortex-hippocampus connectivity and impairs episodic memory [37], demonstrating the feasibility of conducting similar studies in children exposed to early neglect and deprivation. Using highresolution dMRI, we found a 10% reduction in structural connectivity between the entorhinal cortex and the dorsal hippocampus, with limited evidence for sexspecific effects (Fig. 7). In contrast, retrograde labeling revealed a 75% reduction in connectivity in LB males but not females (Figs. 8, 9), highlighting important differences in sensitivity between the two approaches. These differences likely reflect the lack of directionality and cell specificity associated with dMRI [78-80] and the need to further refine tractography tools in the mouse [58, 59, 81].

LB impairs myelination in the SLM

Abnormal myelination is one of the most consistent findings in individuals exposed to early deprivation, with similar findings reported in nonhuman primates and rodents [82]. Only a few studies have examined the effects of LB on myelin development [47, 83], and the mechanisms responsible for these changes are yet to be clarified. Here, we show that LB inhibits OPC differentiation and myelination in the SLM of P17 prepubescent pups, with males being more impacted. This blockade is partly due to reduced axonal innervation, but the presence of hypomyelinated axons in the SLM of P17 males (Fig. 4) suggests that other factors also contributed to reduced myelination. Since myelination is driven by neuronal activation [82], it is possible that the impoverished conditions associated with LB cause abnormally low levels of neuronal activation in perforant pathway terminals that further impair OPC differentiation. In addition, recent work has shown that LB reduces the expression of sodium channels in OPC located in the developing hippocampus [83]. These channels may play a role in promoting OPC differentiation in response to neuronal activity [84–86]. Changes in microglial function may also contribute to the blockade of OPC differentiation. For example, LB reduces the expression of the receptor TREM2 in microglia [48], and low TREM2 levels are associated with abnormal OPC differentiation in a cuprizone model of remyelination [87]. LB also increases TNF α levels in microglia,

which has been shown to inhibit OPC differentiation [88, 89]. The numbers of OPCs and mature oligodendrocytes in the SLM were similar in adolescent P34 CTL and LB mice indicating that the blockade of OPC differentiation is transient. Nevertheless, levels of MBP staining in the SLM continued to be lower in adolescent LB mice (Additional file 1: Fig S5), suggesting that deficits in myelination persist at this age. Whether hypomyelination reflects a reduction in axonal availability or other mechanisms are questions that remain unresolved. Additional studies are needed to clarify whether agents that augment myelination [82] could recover myelination in the SLM and improve contextual fear learning in LB mice.

Perspectives and significance

This work bolsters earlier studies demonstrating that LB causes more pronounced hippocampus-dependent deficits in male compared to female mice. Importantly, we found that cognitive abnormalities are highly correlated with reduced connectivity between the LEC and the dorsal hippocampus. Given that these connections are essential for normal hippocampal function, these deficits in connectivity provide a novel and compelling mechanism to explain why males show more severe impairment in contextual fear conditioning. Adolescent LB male and female mice are hyperactive and have extensive cortical atrophy consistent with outcomes reported in children and adolescents exposed to severe early deprivation. These observations suggest that LB is a mouse model of early deprivation and that many developmental outcomes are similarly impacted in males and females. Elucidating the developmental principles by which ELA establishes sexual dimorphic changes in some circuits but not in others is an important area of investigation for future studies.

Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s13293-024-00616-0.

Additional file1 (DOCX 9783 KB)

Acknowledgements

We want to thank Drs. Cumberbatch, Jamwal, and MacDowell-Kaswan for helpful comments on the manuscript and the Yale Center for Genomic for technical assistance and analysis of the RNA sequencing.

Author contributions

RI- designed the experiments and collected and analyzed the data for Figs. 1– 4, 6, 8, 9. JDW- collected and analyzed data for RNA-seq and supplemental Fig. 1. TMA and JZ- conducted the dMRI studies, including imaging the tissue and analysis (Figs. 5, 7). SM- conducted the RNA-seq analysis. XL-processed tissue and collected images for electron microcopy. BP- ran the contextual fear conditioning and helped with limited bedding studies and tissue collection. LG- helped with the limited bedding studies, tissue collection, immunohistochemistry, and microscopy. CB and SA helped with the limited bedding studies and tissue collection. AK- designed the experiments, helped with data analysis, and wrote the paper. All authors contributed to data interpretation, provided comments, and edited and approved the final version of the manuscript.

Funding

This work was supported by NIMH R01MH119164, NIMH R01MH118332, and the Clinical Neuroscience Division of the VA National Center for PTSD.

Availability of data and materials

RNA-seq raw data and analyses are available at GEO archive accession # GSE142305. All dMRI scans will be available at Figshare.

Declarations

Ethics approval and consent to participate

See Methods (Animals section).

Consent for publication

Not applicable.

Competing interests

The authors have no competing interests to declare.

Author details

¹ Department of Psychiatry, Yale University School of Medicine, 300 George Street, Suite 901, New Haven, CT 06511, USA. ²Bernard and Irene Schwartz Center for Biomedical Imaging, Department of Radiology, New York University School of Medicine, New York, NY 10016, USA. ³Department of Biomedical Engineering, Center for Neurotechnology in Mental Health Research (CNMHR), The Pennsylvania State University, University Park, PA 16802, USA. ⁴Yale Center for Genomic Analysis, P.O. Box 27386, West Haven, CT 06516-7386, USA. ⁵Department of Cell Biology, Yale University School of Medicine, 333 Cedar Street, SHM IE26, New Haven, CT 06510, USA. ⁶Center for Cellular and Molecular Imaging, Electron Microscopy Core Facility, Yale University School of Medicine, New Haven, CT, USA.

Received: 7 November 2023 Accepted: 26 April 2024 Published online: 07 May 2024

References

- White JD, Kaffman A. The moderating effects of sex on consequences of childhood maltreatment: from clinical studies to animal models. Front Neurosci. 2019;13:1082. https://doi.org/10.3389/fnins.2019.01082.
- Teicher MH, Samson JA. Annual research review: enduring neurobiological effects of childhood abuse and neglect. J Child Psychol Psychiatry. 2016;57(3):241–66. https://doi.org/10.1111/jcpp.12507.
- McLaughlin KA, Sheridan MA, Lambert HK. Childhood adversity and neural development: deprivation and threat as distinct dimensions of early experience. Neurosci Biobehav Rev. 2014;47:578–91. https://doi.org/ 10.1016/j.neubiorev.2014.10.012.
- Keyes KM, Eaton NR, Krueger RF, McLaughlin KA, Wall MM, Grant BF, et al. Childhood maltreatment and the structure of common psychiatric disorders. Br J Psychiatry. 2012;200(2):107–15. https://doi.org/10.1192/bjp. bp.111.093062.
- Kessler RC, Davis CG, Kendler KS. Childhood adversity and adult psychiatric disorder in the US National Comorbidity Survey. Psychol Med. 1997;27(5):1101–19.
- Anda RF, Felitti VJ, Bremner JD, Walker JD, Whitfield C, Perry BD, et al. The enduring effects of abuse and related adverse experiences in childhood. A convergence of evidence from neurobiology and epidemiology. Eur Arch Psychiatry Clin Neurosci. 2006;256(3):174–86. https://doi.org/10. 1007/s00406-005-0624-4.
- McLaughlin KA, Colich NL, Rodman AM, Weissman DG. Mechanisms linking childhood trauma exposure and psychopathology: a transdiagnostic model of risk and resilience. BMC Med. 2020;18(1):96. https://doi.org/10. 1186/s12916-020-01561-6.

- Lambert HK, Peverill M, Sambrook KA, Rosen ML, Sheridan MA, McLaughlin KA. Altered development of hippocampus-dependent associative learning following early-life adversity. Dev Cogn Neurosci. 2019;38: 100666. https://doi.org/10.1016/j.dcn.2019.100666.
- Crozier JC, Wang L, Huettel SA, De Bellis MD. Neural correlates of cognitive and affective processing in maltreated youth with posttraumatic stress symptoms: does gender matter? Dev Psychopathol. 2014;26(2):491–513. https://doi.org/10.1017/S095457941400008X.
- Lambert HK, Sheridan MA, Sambrook KA, Rosen ML, Askren MK, McLaughlin KA. Hippocampal contribution to context encoding across development is disrupted following early-life adversity. J Neurosci. 2017;37(7):1925–34. https://doi.org/10.1523/JNEUROSCI.2618-16.2017.
- De Bellis MD, Woolley DP, Hooper SR. Neuropsychological findings in pediatric maltreatment: relationship of PTSD, dissociative symptoms, and abuse/neglect indices to neurocognitive outcomes. Child Maltreat. 2013;18(3):171–83. https://doi.org/10.1177/1077559513497420.
- Gershon A, Minor K, Hayward C. Gender, victimization, and psychiatric outcomes. Psychol Med. 2008;38(10):1377–91. https://doi.org/10.1017/ S0033291708003000.
- Khalaf-Nazzal R, Francis F. Hippocampal development old and new findings. Neuroscience. 2013;248:225–42. https://doi.org/10.1016/j.neuro science.2013.05.061.
- 14. Li G, Pleasure SJ. Morphogenesis of the dentate gyrus: what we are learning from mouse mutants. Dev Neurosci. 2005;27(2–4):93–9.
- Strange BA, Witter MP, Lein ES, Moser EI. Functional organization of the hippocampal longitudinal axis. Nat Rev Neurosci. 2014;15(10):655–69. https://doi.org/10.1038/nrn3785.
- Rocha M, Wang D, Avila-Quintero V, Bloch MH, Kaffman A. Deficits in hippocampal-dependent memory across different rodent models of early life stress: systematic review and meta-analysis. Transl Psychiatry. 2021;11(1):231. https://doi.org/10.1038/s41398-021-01352-4.
- Chen Y, Kramar EA, Chen LY, Babayan AH, Andres AL, Gall CM, et al. Impairment of synaptic plasticity by the stress mediator CRH involves selective destruction of thin dendritic spines via RhoA signaling. Mol Psychiatry. 2013;18(4):485–96. https://doi.org/10.1038/mp.2012.17.
- Bolton JL, Molet J, Ivy A, Baram TZ. New insights into early-life stress and behavioral outcomes. Curr Opin Behav Sci. 2017;14:133–9. https://doi. org/10.1016/j.cobeha.2016.12.012.
- Bolton JL, Schulmann A, Garcia-Curran MM, Regev L, Chen Y, Kamei N, et al. Unexpected transcriptional programs contribute to hippocampal memory deficits and neuronal stunting after early-life adversity. Cell Rep. 2020;33(11):108511. https://doi.org/10.1016/j.celrep.2020.108511.
- Naninck EF, Hoeijmakers L, Kakava-Georgiadou N, Meesters A, Lazic SE, Lucassen PJ, et al. Chronic early life stress alters developmental and adult neurogenesis and impairs cognitive function in mice. Hippocampus. 2015;25(3):309–28. https://doi.org/10.1002/hipo.22374.
- Bath KG, Nitenson AS, Lichtman E, Lopez C, Chen W, Gallo M, et al. Early life stress leads to developmental and sex selective effects on performance in a novel object placement task. Neurobiol Stress. 2017;7:57–67. https://doi.org/10.1016/j.ynstr.2017.04.001.
- Youssef M, Atsak P, Cardenas J, Kosmidis S, Leonardo ED, Dranovsky A. Early life stress delays hippocampal development and diminishes the adult stem cell pool in mice. Sci Rep. 2019;9(1):4120. https://doi.org/10. 1038/s41598-019-40868-0.
- Walker CD, Bath KG, Joels M, Korosi A, Larauche M, Lucassen PJ, et al. Chronic early life stress induced by limited bedding and nesting (LBN) material in rodents: critical considerations of methodology, outcomes and translational potential. Stress. 2017;20(5):421–48. https://doi.org/10. 1080/10253890.2017.1343296.
- Seress L. Comparative anatomy of the hippocampal dentate gyrus in adult and developing rodents, non-human primates and humans. Prog Brain Res. 2007;163:23–41. https://doi.org/10.1016/S0079-6123(07) 63002-7.
- Workman AD, Charvet CJ, Clancy B, Darlington RB, Finlay BL. Modeling transformations of neurodevelopmental sequences across mammalian species. J Neurosci. 2013;33(17):7368–83. https://doi.org/10.1523/JNEUR OSCI.5746-12.2013.
- Zeiss CJ. Comparative milestones in rodent and human postnatal central nervous system development. Toxicol Pathol. 2021;49(8):1368–73. https:// doi.org/10.1177/01926233211046933.

- Abraham H, Vincze A, Jewgenow I, Veszpremi B, Kravjak A, Gomori E, et al. Myelination in the human hippocampal formation from midgestation to adulthood. Int J Dev Neurosci. 2010;28(5):401–10. https://doi. org/10.1016/j.ijdevneu.2010.03.004.
- Lavenex P, Banta LP. Building hippocampal circuits to learn and remember: ber: insights into the development of human memory. Behav Brain Res. 2013;254:8–21. https://doi.org/10.1016/j.bbr.2013.02.007.
- Lynch KM, Shi Y, Toga AW, Clark KA, Pediatric Imaging N, Genetics S. Hippocampal shape maturation in childhood and adolescence. Cereb Cortex. 2019;29(9):3651–65. https://doi.org/10.1093/cercor/bhy244.
- Nilssen ES, Doan TP, Nigro MJ, Ohara S, Witter MP. Neurons and networks in the entorhinal cortex: a reappraisal of the lateral and medial entorhinal subdivisions mediating parallel cortical pathways. Hippocampus. 2019;29(12):1238–54. https://doi.org/10.1002/hipo.23145.
- Super H, Soriano E. The organization of the embryonic and early postnatal murine hippocampus. II. Development of entorhinal, commissural, and septal connections studied with the lipophilic tracer Dil. J Comp Neurol. 1994;344(1):101–20. https://doi.org/10.1002/cne.90344 0108.
- Deng JB, Yu DM, Li MS. Formation of the entorhino-hippocampal pathway: a tracing study in vitro and in vivo. Neurosci Bull. 2006;22(6):305–14.
- Amaral DG, Kondo H, Lavenex P. An analysis of entorhinal cortex projections to the dentate gyrus, hippocampus, and subiculum of the neonatal macaque monkey. J Comp Neurol. 2014;522(7):1485–505. https://doi.org/ 10.1002/cne.23469.
- Woods NI, Vaaga CE, Chatzi C, Adelson JD, Collie MF, Perederiy JV, et al. Preferential targeting of lateral entorhinal inputs onto newly integrated granule cells. J Neurosci. 2018;38(26):5843–53. https://doi.org/10.1523/ JNEUROSCI.1737-17.2018.
- Vandrey B, Garden DLF, Ambrozova V, McClure C, Nolan MF, Ainge JA. Fan cells in layer 2 of the lateral entorhinal cortex are critical for episodic-like memory. Curr Biol. 2020;30(1):169-75 e5. https://doi.org/10.1016/j.cub. 2019.11.027.
- Xu X, Song L, Kringel R, Hanganu-Opatz IL. Developmental decrease of entorhinal-hippocampal communication in immune-challenged DISC1 knockdown mice. Nat Commun. 2021;1(1):6810. https://doi.org/10.1038/ s41467-021-27114-w.
- Wilde EA, Hyseni I, Lindsey HM, Faber J, McHenry JM, Bigler ED, et al. A preliminary DTI tractography study of developmental neuroplasticity 5–15 years after early childhood traumatic brain injury. Front Neurol. 2021;12: 734055. https://doi.org/10.3389/fneur.2021.734055.
- Zhao R, Grunke SD, Keralapurath MM, Yetman MJ, Lam A, Lee TC, et al. Impaired recall of positional memory following chemogenetic disruption of place field stability. Cell Rep. 2016;16(3):793–804. https://doi.org/10. 1016/j.celrep.2016.06.032.
- Clarkson J, Herbison AE. Postnatal development of kisspeptin neurons in mouse hypothalamus; sexual dimorphism and projections to gonadotropin-releasing hormone neurons. Endocrinology. 2006;147(12):5817–25.
- Roa J, Aguilar E, Dieguez C, Pinilla L, Tena-Sempere M. New frontiers in kisspeptin/GPR54 physiology as fundamental gatekeepers of reproductive function. Front Neuroendocrinol. 2008;29(1):48–69.
- d'Anglemont de Tassigny X, Fagg LA, Dixon JP, Day K, Leitch HG, Hendrick AG, et al. Hypogonadotropic hypogonadism in mice lacking a functional Kiss1 gene. Proc Natl Acad Sci USA. 2007;104(25):10714–9. https://doi. org/10.1073/pnas.0704114104.
- Michael SD, Kaplan SB, Macmillan BT. Peripheral plasma concentrations of LH, FSH, prolactin and GH from birth to puberty in male and female mice. J Reprod Fertil. 1980;59(1):217–22. https://doi.org/10.1530/jrf.0.0590217.
- Drickamer LC. Contact stimulation, androgenized females and accelerated sexual maturation in female mice. Behav Biol. 1974;12(1):101–10.
- 44. Li Y, Mu Y, Gage FH. Development of neural circuits in the adult hippocampus. Curr Top Dev Biol. 2009;87:149–74.
- Lemasson M, Saghatelyan A, Olivo-Marin JC, Lledo PM. Neonatal and adult neurogenesis provide two distinct populations of newborn neurons to the mouse olfactory bulb. J Neurosci. 2005;25(29):6816–25.
- Nicola Z, Fabel K, Kempermann G. Development of the adult neurogenic niche in the hippocampus of mice. Front Neuroanat. 2015;9:53. https:// doi.org/10.3389/fnana.2015.00053.
- 47. Johnson FK, Delpech JC, Thompson GJ, Wei L, Hao J, Herman P, et al. Amygdala hyper-connectivity in a mouse model of unpredictable

early life stress. Transl Psychiatry. 2018;8(1):49. https://doi.org/10.1038/s41398-018-0092-z.

- Dayananda KK, Ahmed S, Wang D, Polis B, Islam R, Kaffman A. Early life stress impairs synaptic pruning in the developing hippocampus. Brain Behav immun. 2022. https://doi.org/10.1016/j.bbi.2022.09.014.
- Frankish A, Diekhans M, Ferreira AM, Johnson R, Jungreis I, Loveland J, et al. GENCODE reference annotation for the human and mouse genomes. Nucleic Acids Res. 2019;47(D1):D766–73. https://doi.org/10. 1093/nar/gky955.
- Kim D, Langmead B, Salzberg SL. HISAT: a fast spliced aligner with low memory requirements. Nat Methods. 2015;12(4):357–60. https://doi. org/10.1038/nmeth.3317.
- Steadman PE, Xia F, Ahmed M, Mocle AJ, Penning ARA, Geraghty AC, et al. Disruption of oligodendrogenesis impairs memory consolidation in adult mice. Neuron. 2020;105(1):150-64 e6. https://doi.org/10.1016/j. neuron.2019.10.013.
- Liu J, Dupree JL, Gacias M, Frawley R, Sikder T, Naik P, et al. Clemastine enhances myelination in the prefrontal cortex and rescues behavioral changes in socially isolated mice. J Neurosci. 2016;36(3):957–62. https://doi.org/10.1523/JNEUROSCI.3608-15.2016.
- McKenzie IA, Ohayon D, Li H, de Faria JP, Emery B, Tohyama K, et al. Motor skill learning requires active central myelination. Science. 2014;346(6207):318–22. https://doi.org/10.1126/science.1254960.
- Munz M, Bharioke A, Kosche G, Moreno-Juan V, Brignall A, Rodrigues TM, et al. Pyramidal neurons form active, transient, multilayered circuits perturbed by autism-associated mutations at the inception of neocortex. Cell. 2023;186(9):1930-49 e31. https://doi.org/10.1016/j.cell.2023. 03.025.
- West MJ. Stereological methods for estimating the total number of neurons and synapses: issues of precision and bias. Trends Neurosci. 1999;22(2):51–61.
- Delpech JC, Wei L, Hao J, Yu X, Madore C, Butovsky O, et al. Early life stress perturbs the maturation of microglia in the developing hippocampus. Brain Behav Immun. 2016. https://doi.org/10.1016/j.bbi. 2016.06.006.
- Kaiser T, Allen HM, Kwon O, Barak B, Wang J, He Z, et al. MyelTracer: a semi-automated software for myelin g-ratio quantification. eNeuro. 2021;8:4. https://doi.org/10.1523/ENEURO.0558-20.2021.
- White JD, Arefin TM, Pugliese A, Lee CH, Gassen J, Zhang J, et al. Early life stress causes sex-specific changes in adult fronto-limbic connectivity that differentially drive learning. Elife. 2020. https://doi.org/10.7554/eLife. 58301.
- Arefin TM, Lee CH, White JD, Zhang J, Kaffman A. Macroscopic structural and connectome mapping of the mouse brain using diffusion magnetic resonance imaging. Bio Protoc. 2021;11(22): e4221. https://doi.org/10. 21769/BioProtoc.4221.
- Arefin TM, Lee CH, Liang Z, Rallapalli H, Wadghiri YZ, Turnbull DH, et al. Towards reliable reconstruction of the mouse brain corticothalamic connectivity using diffusion MRI. Neuroimage. 2023;273: 120111. https://doi. org/10.1016/j.neuroimage.2023.120111.
- 61. Love MI, Huber W, Anders S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. Genome Biol. 2014;15(12):550. https://doi.org/10.1186/s13059-014-0550-8.
- Bujalka H, Koenning M, Jackson S, Perreau VM, Pope B, Hay CM, et al. MYRF is a membrane-associated transcription factor that autoproteolytically cleaves to directly activate myelin genes. PLoS Biol. 2013;11(8): e1001625. https://doi.org/10.1371/journal.pbio.1001625.
- Meier S, Brauer AU, Heimrich B, Nitsch R, Savaskan NE. Myelination in the hippocampus during development and following lesion. Cell Mol Life Sci. 2004;61(9):1082–94. https://doi.org/10.1007/s00018-004-3469-5.
- 64. Tiwari SK, Agarwal S, Chauhan LK, Mishra VN, Chaturvedi RK. Bisphenol-A impairs myelination potential during development in the hippocampus of the rat brain. Mol Neurobiol. 2015;51(3):1395–416. https://doi.org/10. 1007/s12035-014-8817-3.
- McLaughlin KA, Sheridan MA, Winter W, Fox NA, Zeanah CH, Nelson CA. Widespread reductions in cortical thickness following severe early-life deprivation: a neurodevelopmental pathway to attention-deficit/hyperactivity disorder. Biol Psychiatry. 2014;76(8):629–38. https://doi.org/10. 1016/j.biopsych.2013.08.016.
- 66. Sheridan MA, Mukerji CE, Wade M, Humphreys KL, Garrisi K, Goel S, et al. Early deprivation alters structural brain development from middle

childhood to adolescence. Sci Adv. 2022;8(40):eabn4316. https://doi.org/ 10.1126/sciadv.abn4316.

- 67. Sarnat HB. Axonal pathfinding during the development of the nervous system. Ann Child Neurol Soc. 2023;1(1):13–23. https://doi.org/10.1002/cns3.2.
- Brinks H, Conrad S, Vogt J, Oldekamp J, Sierra A, Deitinghoff L, et al. The repulsive guidance molecule RGMa is involved in the formation of afferent connections in the dentate gyrus. J Neurosci. 2004;24(15):3862–9. https://doi.org/10.1523/JNEUROSCI.5296-03.2004.
- 69. Saxena S, Caroni P. Mechanisms of axon degeneration: from development to disease. Prog Neurobiol. 2007;83(3):174–91. https://doi.org/10.1016/j. pneurobio.2007.07.007.
- Ahmed S, Polis B, Jamwal S, Sanganahalli BG, MacDowell Kaswan Z, Islam R, et al. Transient impairment in microglial function causes sex-specific deficits in synaptic and hippocampal function in mice exposed to early adversity. bioRxiv. 2024. https://doi.org/10.1101/2024.02.14.580284.
- Sedlak AJ, Mettenburg J, Basena M, Petta I, McPherson K, Greene A, et al. Fourth national incidence study of child abuse and neglect (NIS–4): report to congress. Washington: US Department of Health and Human Services, Administration for Children and Families; 2010.
- Child Maltreatment. Child abuse, neglect data released, 29th edition of the Child Maltreatment Report. https://www.acf.hhs.gov/cb/report/childmaltreatment-2022.
- McLaughlin KA, Sheridan MA, Nelson CA. Neglect as a violation of species-expectant experience: neurodevelopmental consequences. Biol Psychiatry. 2017;82(7):462–71. https://doi.org/10.1016/j.biopsych.2017.02. 1096.
- De Bellis MD. The psychobiology of neglect. Child Maltreat. 2005;10(2):150–72. https://doi.org/10.1177/1077559505275116.
- Mackes NK, Golm D, Sarkar S, Kumsta R, Rutter M, Fairchild G, et al. Early childhood deprivation is associated with alterations in adult brain structure despite subsequent environmental enrichment. Proc Natl Acad Sci USA. 2020;117(1):641–9. https://doi.org/10.1073/pnas.1911264116.
- Bos KJ, Fox N, Zeanah CH, Nelson Iii CA. Effects of early psychosocial deprivation on the development of memory and executive function. Front Behav Neurosci. 2009;3:16. https://doi.org/10.3389/neuro.08.016.2009.
- Terock J, Van der Auwera S, Janowitz D, Wittfeld K, Frenzel S, Klinger-Konig J, et al. Childhood trauma and adult declarative memory performance in the general population: the mediating effect of alexithymia. Child Abuse Negl. 2020;101: 104311. https://doi.org/10.1016/j.chiabu.2019.104311.
- Maier-Hein KH, Neher PF, Houde JC, Cote MA, Garyfallidis E, Zhong J, et al. The challenge of mapping the human connectome based on diffusion tractography. Nat Commun. 2017;8(1):1349. https://doi.org/10.1038/ s41467-017-01285-x.
- Pallast N, Wieters F, Nill M, Fink GR, Aswendt M. Graph theoretical quantification of white matter reorganization after cortical stroke in mice. Neuroimage. 2020;217: 116873. https://doi.org/10.1016/j.neuroimage. 2020.116873.
- Lerch JP, van der Kouwe AJ, Raznahan A, Paus T, Johansen-Berg H, Miller KL, et al. Studying neuroanatomy using MRI. Nat Neurosci. 2017;20(3):314–26. https://doi.org/10.1038/nn.4501.
- Muller HP, Roselli F, Rasche V, Kassubek J. Diffusion tensor imaging-based studies at the group-level applied to animal models of neurodegenerative diseases. Front Neurosci. 2020;14:734. https://doi.org/10.3389/fnins. 2020.00734.
- Islam R, Kaffman A. White-matter repair as a novel therapeutic target for early adversity. Front neurosci. 2021. https://doi.org/10.3389/fnins.2021. 657693.
- Treccani G, Yigit H, Lingner T, Schleubetaner V, Mey F, van der Kooij MA, et al. Early life adversity targets the transcriptional signature of hippocampal NG2+ glia and affects voltage gated sodium (Na(v)) channels properties. Neurobiol Stress. 2021;15: 100338. https://doi.org/10.1016/j. ynstr.2021.100338.
- Paez PM, Fulton D, Colwell CS, Campagnoni AT. Voltage-operated Ca(2+) and Na(+) channels in the oligodendrocyte lineage. J Neurosci Res. 2009;87(15):3259–66. https://doi.org/10.1002/jnr.21938.
- Karadottir R, Hamilton NB, Bakiri Y, Attwell D. Spiking and nonspiking classes of oligodendrocyte precursor glia in CNS white matter. Nat Neurosci. 2008;11(4):450–6. https://doi.org/10.1038/nn2060.
- Spitzer SO, Sitnikov S, Kamen Y, Evans KA, Kronenberg-Versteeg D, Dietmann S, et al. Oligodendrocyte progenitor cells become regionally

diverse and heterogeneous with age. Neuron. 2019;101(3):459-71 e5. https://doi.org/10.1016/j.neuron.2018.12.020.

- Cignarella F, Filipello F, Bollman B, Cantoni C, Locca A, Mikesell R, et al. TREM2 activation on microglia promotes myelin debris clearance and remyelination in a model of multiple sclerosis. Acta Neuropathol. 2020;140(4):513–34. https://doi.org/10.1007/s00401-020-02193-z.
- Bonora M, De Marchi E, Patergnani S, Suski JM, Celsi F, Bononi A, et al. Tumor necrosis factor-alpha impairs oligodendroglial differentiation through a mitochondria-dependent process. Cell Death Differ. 2014;21(8):1198–208. https://doi.org/10.1038/cdd.2014.35.
- Pang Y, Zheng B, Fan LW, Rhodes PG, Cai Z. IGF-1 protects oligodendrocyte progenitors against TNFalpha-induced damage by activation of PI3K/Akt and interruption of the mitochondrial apoptotic pathway. Glia. 2007;55(11):1099–107. https://doi.org/10.1002/glia.20530.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.